Check for updates

OPEN ACCESS

EDITED BY Sanjat Kanjilal, Harvard Pilgrim Health Care and Harvard Medical School, United States

REVIEWED BY Ming Huang, Nagoya City University, Japan Tanmoy Sarkar Pias, Virginia Tech, United States

*CORRESPONDENCE Constantinos S. Pattichis ⊠ pattichi@ucy.ac.cy

RECEIVED 19 September 2024 ACCEPTED 16 December 2024 PUBLISHED 12 February 2025

CITATION

Sulaiman WA, Stylianides C, Nikolaou A, Antoniou Z, Constantinou I, Palazis L, Vavlitou A, Kyprianou T, Kyriacou E, Kakas A, Pattichis MS, Panayides AS and Pattichis CS (2025) Leveraging machine learning and rule extraction for enhanced transparency in emergency department length of stay prediction.

Front. Digit. Health 6:1498939. doi: 10.3389/fdgth.2024.1498939

COPYRIGHT

© 2025 Sulaiman, Stylianides, Nikolaou, Antoniou, Constantinou, Palazis, Vavlitou, Kyprianou, Kyriacou, Kakas, Pattichis, Panayides and Pattichis. This is an openaccess article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Leveraging machine learning and rule extraction for enhanced transparency in emergency department length of stay prediction

Waqar A. Sulaiman¹, Charithea Stylianides², Andria Nikolaou^{1,2}, Zinonas Antoniou³, Ioannis Constantinou³, Lakis Palazis⁴, Anna Vavlitou⁴, Theodoros Kyprianou^{5,6}, Efthyvoulos Kyriacou⁷, Antonis Kakas^{1,2}, Marios S. Pattichis⁸, Andreas S. Panayides² and Constantinos S. Pattichis^{1,2*}

¹Department of Computer Science and Biomedical Engineering Research Centre, University of Cyprus, Nicosia, Cyprus, ²CYENS Centre of Excellence, Nicosia, Cyprus, ³Research & Development Department, 3AHealth, Nicosia, Cyprus, ⁴Department of Intensive Care Medicine, Limassol General Hospital, State Health Services Organisation, Nicosia, Cyprus, ⁵Department of Critical Care and Emergency Medicine, Medical School, University of Nicosia, Nicosia, Cyprus, ⁶Department of Critical Care, St Thomas's Hospital NHS, London, United Kingdom, ⁷Department of Electrical Engineering, Computer Engineering and Informatics, Cyprus University of Technology, Limassol, Cyprus, ⁸Department of Electrical and Computer Engineering, University of New Mexico, Albuquerque, NM, United States

This study aims to address the critical issue of emergency department (ED) overcrowding, which negatively affects patient outcomes, wait times, and resource efficiency. Accurate prediction of ED length of stay (LOS) can streamline operations and improve care delivery. We utilized the MIMIC IV-ED dataset, comprising over 400,000 patient records, to classify ED LOS into short (\leq 4.5 hours) and long (>4.5 hours) categories. Using machine learning models, including Gradient Boosting (GB), Random Forest (RF), Logistic Regression (LR), and Multilayer Perceptron (MLP), we identified GB as the best performing model outperforming the other models with an AUC of 0.730, accuracy of 69.93%, sensitivity of 88.20%, and specificity of 40.95% on the original dataset. In the balanced dataset, GB had an AUC of 0.729, accuracy of 68.86%, sensitivity of 75.39%, and specificity of 58.59%. To enhance interpretability, a novel rule extraction method for GB model was implemented using relevant important predictors, such as triage acuity, comorbidity scores, and arrival methods. By combining predictive analytics with interpretable rule-based methods, this research provides actionable insights for optimizing patient flow and resource allocation. The findings highlight the importance of transparency in machine learning applications for healthcare, paving the way for future improvements in model performance and clinical adoption.

KEYWORDS

emergency department, length of stay, machine learning, gradient boosting, rule extraction, predictive modeling, explainable AI, healthcare analytics

1 Introduction

Emergency departments (EDs) are at the forefront of healthcare, where providing timely and efficient care is essential for saving lives. Overcrowding is a global problem that causes longer wait times, lower quality care, and increased stress for healthcare providers (1). Overcrowding occurs when the demand for emergency services exceeds a department's capacity to provide timely, high-quality care, negatively affecting patient outcomes and operational efficiency (2).

Effective management of emergency department (ED) length of stay (LOS) is crucial to mitigating overcrowding, primarily by optimizing the allocation of resource (3). Accurate prediction of LOS can help streamline patient flow, prioritize treatment, and ensure that resources are used efficiently, resulting in better patient outcomes in the ED. Automating ED LOS classification with machine learning (ML) techniques allows healthcare providers to make proactive data-driven decisions, especially in high-volume EDs where traditional methods are not scalable or accurate (4).

Traditional LOS prediction methods often use simplistic metrics or heuristic rules, which do not account for the complexity of patient conditions and clinical pathways (4). These approaches introduce biases and overlook important patientspecific factors, reducing resource planning effectiveness and increasing the likelihood of suboptimal care delivery, particularly under high-pressure conditions typical of EDs.

Recent advances in machine learning have shown promise in improving the accuracy and granularity of ED LOS classification (5, 6). However, many ML models are considered black boxes, which limits their practical application in clinical settings due to difficulties in understanding the underlying decision processes (7–9). To address this issue, our study aims not only to accurately classify ED LOS using the MIMIC IV ED healthcare dataset, but also to improve model interpretability by implementing a novel rule extraction method tailored for Gradient Boosting (GB).

We use machine learning and deep learning models such as GB, Random Forest (RF), Logistic Regression (LR), and Multilayer Preceptron (MLP) to categorize ED LOS into short stays (\leq 4.5 h) and long stays (>4.5 h), identifying the model with the best predictive capabilities. In addition, our goal is to improve model transparency by analyzing the extracted rules and calculating metrics such as accuracy, relative coverage, and overall coverage. This rigorous analysis provides healthcare professionals with clear and actionable guidelines for patient management, facilitating informed clinical decision-making, and optimal allocation of resources in emergency departments (10, 11).

Our research also addresses the class imbalance of the data set, which shows that "ED LOS (Long)" cases outnumber "ED LOS (Short)" cases. To address this imbalance, we employ various different types of over- and under-sampling methods in a robust manner, enhancing the model's ability to learn from both classes effectively (12).

By validating these findings and exploring the integration of sophisticated predictive models with interpretable rule extraction techniques, this research aims to optimize patient flow and refine care delivery strategies in emergency healthcare settings. This approach contributes to the advancement of data-driven methodologies in healthcare management, illustrating how ML techniques can enhance operational efficiency and patient outcomes.

The structure of the paper is organized as follows: Section 2 provides an overview of studies on Emergency Department Length of Stay. Section 3 details the materials and methods, including feature selection, machine learning modeling, and rule extraction. Section 4 presents the results of the feature analysis, model evaluation, and rule extraction. Section 5 discusses these findings, and finally Section 6 concludes the article. A complete overview of the study flow is presented in Figure 1.



2 An overview of emergency department length of stay studies

Recent investigations into predicting emergency department length of stay have utilized various machine learning models to meet the growing demand for effective patient care and resource management. The objectives, features, datasets, and methodologies of these studies vary, as shown in Table 1. This section provides a comprehensive overview of current research in this area.

Ala et al. (13) and Zebin et al. (14). Both used the MIMIC data set, although different versions, to investigate patient outcomes in emergency settings. Ala et al. (13) classified emergency patients into complex and fast-track categories using models such as Multilayer Perceptron (MLP), Regression Trees (CART), and Stochastic Gradient Boosting (SGB), with the SGB model reaching an AUC of 0.752. This study demonstrated the importance of clinical variables, such as vital signs and chief complaints, in designing successful decision support systems. Zebin et al (14) used an auto-encoded deep neural network on the MIMIC-III dataset to differentiate between short and long hospital stays. Their solution, which obtained an accuracy of 77.7%, demonstrated the value of advanced feature engineering and deep learning to capture complex patterns in large datasets. Both studies emphasize the need to integrate diverse patient data to improve prediction accuracy and clinical decision-making.

Pedro et al. (15) used a ProAGE (Pronto Atendimento Geritrico Especializado – Specialized Geriatric Emergency Care) rating system which is a validated mnemonic method used to assess vulnerability and predict hospital admission, prolonged length of stay (LoS), and death in older adults at the ED (21). They combined ProAGE with bootstrapping to predict hospital admission and prolonged LOS in older people, achieving an AUC of 0.790. Their study focused on a specific cohort, emphasizing the importance of geriatric-specific characteristics

TABLE 1 Summary of studies on ED LOS prediction.

Study	Objective	Features	Dataset	Models	Sample size and train/test split	Results
Ala et al. (13)	Classify emergency patients into complex and fast-track (<4 h)	19 variables which include vital signs, pain, chief complaints, arrival means, and age	MIMIC-IV-ED and MIMIC-IV	MLP, CART, RF, LR, SGB	sample size = 104,014, Split: 80% train, 20% test, 5-folds CV	SGB: Sensitivity = 59.13%, Specificity = 91.20%, AUC = 0.752
Zebin et al. (14)	Identify short (0–7 d) and long (>7 d) hospital stays	Admission records, demographics, diagnosis codes (ICD-9), chart events	MIMIC-III	RF, Auto- Encoder, DNN	Sample Size: 53,104 Split: 80% train, 10% validation, 10% test	Auto+DNN: Accuracy = 77.7%, Sensitivity = 77.6%, Precision = 75.2%
Pedro et al. (15)	Predict hospital admission and prolonged LOS in older adults (>70 years)	21 variables including age, sex, comorbidities, ISAR tool, FRAIL scale, and CAM	Retrospective cohort of geriatric ED in Brazil	PRO-AGE scoring system, Bootstrapping	Sample Size: 5,025 patients, Data Split: 2:1 (bootstrapped)	AUC = 0.79
Bopche et al. (16)	Predict prolonged ED LOS (>2 d) using historical patient records	Age, temporal features, laboratory results, diagnostic codes	Retrospective cohort at St Olavs University Hospital, Norway	XGBoost, LightGBM, CatBoost	Sample Size: 35,591 Split: 80/20	XGBoost: AUROC = 0.9797, Accuracy = 0.9286, Sensitivity = 0.9179, Specificty = 0.9322
Etu et al. (17)	Prediction model for COVID-19 patient ED LOS less than or greater than 4 h	127 clinical and operational variables including age, sex, ESI, insurance, and comorbidities	Henry Ford Hospital EHR	LR, GB, DT, RF,	Sample Size: 57,665 Split: 80% train, 20% test, SMOTE, 10-fold cross- validation	GB: Accuracy = 85%, AUC = 0.930, F1 Score = 0.880,
Chang et al. (18)	Predict low-severity patients with short discharge LOS in ED (<4 h)	32 variables including trauma, chief complaints, injury mechanisms, and vital signs	Retrospective study in Taiwan	CatBoost, XGBoost, DT, RF, LR	Sample Size = 127,749, Split: 80% train, 20% test	CatBoost: AUC = 0.748, Sensitivity = 59.09%, Specificity = 83.25%
Rahman et al. (19)	Predict ED LOS >4 h using a decision tree algorithm	33 variables including ED visit type, age, gender, triage category, and consultations	Regional Australian public hospital	Decision Tree	Sample Size: 80,512 Split: divided into mutually exclusive horizontal segments, 10-fold cross- validation	Accuracy = 84.94%
Gill et al. (20)	Predict fast track patients staying longer than 4 h in ED	27 variables with emphasis on time-dependent factors like time to imaging request	Regional Australian Public Hospital	Gradient Boosting	Sample Size: 62,955 Split: 70% train, 30% test, 10-fold cross-validation	AUC = 0.890
This study	Predict ED LOS >4.5 h with enhanced interpretability using rule extraction	Top 10 important variables including arrival methods and medication history	MIMIC IV-ED	GB, RF, LR, SVM,	Sample Size = 410,927, Split: 80% train, 20% test, SMOTE +Tomek, 5-fold stratified cross-validation	GB: Accuracy = 69.93%, Sensitivity = 88.20%, Specificity = 40.95%, AUC = 0.730

CatBoost, Categorical boosting; CART, Classification and regression trees; DNN, Deep neural network; DT, Decision tree; GB, Gradient boosting; LR, Logistic regression; MLP, Multilayer perceptron; ProAGE, Pronto atendimento geriátrico especializado-specialized geriatric emergency care; RF, Random forest; SGB, Stochastic gradient boosting; SMOTE + Tomek, Synthetic minority over-sampling technique + tomeklinks undersampling technique; XGBoost, Extreme gradient boosting.

such as the ISAR (Identification of Seniors At Risk) tool and the FRAIL scale (Fatigue, Resistance, Ambulation, Illness, and Loss of weight), which are critical to understanding the complexities of care for elderly patients.

Etu et al. (17) created a prediction model for COVID-19 patient ED LOS by combining Logistic Regression (LR), Gradient Boosting (GB), and Decision Trees. Their study, which used data from Henry Ford Hospital, found that the GB models outperformed other models, with an AUC of 0.930, especially when dealing with unbalanced data sets using approaches such as SMOTE.

Chang et al. (18) used a variety of models, including CatBoost and XGBoost, to predict low-severity patients with short discharge times. Their work revealed the usefulness of ensemble methods in a retrospective Taiwanese data set, with CatBoost reaching an AUC of 0.748. This study emphasizes the necessity of model selection in different clinical contexts.

Rahman et al. (19) and Gill et al. (20) both studied regional Australian public hospitals, with an emphasis on predicting ED LOS of more than 4 h. Rahman et al. (19) used a decision tree approach and obtained an accuracy of 84.94%. Their research revealed the utility of decision trees in developing interpretable models for the prediction of LOS in the emergency department, which is especially significant in clinical settings where transparency is required. Similarly, Gill et al. (20) focused on fast-track patients and applied gradient boosting, reaching an AUC of 0.890. Their results highlighted the importance of timedependent variables, such as the time to imaging request, in understanding patient flow and addressing bottlenecks in emergency care. Both findings emphasize the importance of predictive modeling in increasing the efficiency of the emergency department and patient outcomes.

Our research builds on previous work by combining rule extraction with GB models to improve interpretability while retaining predictive performance. Our research not only coincides with the accuracy and AUC scores published in previous studies but also makes a distinct addition by offering clear, actionable insights through rule extraction.

3 Materials and methods

In this section, we first describe raw data processing/ benchmark data generation, and preprocessing. Next, we introduce data balancing methods, baseline models for the benchmark task, and model performance evaluation. Finally, we elaborate on the rule extraction and analysis methods. A complete overview of the study flow is presented in Figure 1.

3.1 Data source

We used real-world data from the MIMIC-IV ED dataset (22), which is a comprehensive clinical database that contains detailed information from the emergency departments of Beth Israel Deaconess Medical Center in Boston, Massachusetts. The analytical data set contained deidentified information from more than 400,000 emergency department visits between 2011 and 2019, such as demographics, comorbidities, laboratory results, vital signs, medications, medical procedures, and clinical results. The raw data was managed using a data generation pipeline (7), and additional data processing was performed with Python version 3.9.7. Additional information about the initial data processing is provided to ensure the transparency and reproducibility of our methods.

3.2 Preprocessing

Raw electronic health record (EHR) data is often unsuitable for model building due to issues such as missing values, outliers, duplicates, and errors arising from system or clerical errors (23). Therefore, effective preprocessing is crucial to ensure optimal machine learning performance.

3.2.1 Exclusion criteria

We addressed these issues by first defining the following exclusion criteria:

- Patients under the age of 18 (n = 168) were excluded
- ED LOS less than 0.5 h (n = 2,264) and more than 24 h (n = 11,896) were excluded.
- Negative values of ED LOS (*n* = 6) that were deemed erroneous were removed.

The criteria were guided by existing literature (7, 24), which shows that patients under the age of 18 should be discarded as their medical conditions and healthcare requirements differ significantly from adults, necessitating different triage and treatment procedures. An ED LOS of 24 to 48 h can be considered prolonged (24). Prolonged LOS is frequently attributed to non-health-related factors such as bed shortages or patients refusing to be discharged, whereas stays of less than 0.5 h are typically associated with minor complications.

3.2.2 Outlier detection

Outlier detection was performed, first by defining values of vital signs as outliers and marking them as missing if they fell outside the plausible physiological range determined by domain knowledge, such as a value less than zero or an oxygen saturation level greater than 100%, using an outlier detection procedure similar to MIMIC-EXTRACT (25). For the detection thresholds, we used thresholds found in Harutyunyan et al. (26). One set of upper and lower thresholds was used to filter outliers, and any value outside of this range was marked as missing. A set of thresholds were introduced to indicate the physiologically valid range. Any value that fell outside of this range was replaced with the nearest valid value, see Table 2.

Second, outlier detection was performed on the outcome variable ED LOS. We used several established methods for this purpose, including the Interquartile Range (IQR), Modified Z-score, and unsupervised learning models like Isolation Forest and Local Outlier Factor (LOF) (27, 28). LOF outperformed other methods as it was able to retain a distribution closest to original distribution as seen in Figure 2. A complete overview of the total missing values for all variables discussed above can be found in Table 2.

TABLE 2 Summary of MIMIC-IV ED observation distribution of Vital Signs and ED LOS.

Variable	Total outliers	Total valid values	Valid range (<mark>26</mark>)
Triage temperature (°C)	23,903	401,178	[26, 45]
Triage heartrate (bpm)	17,097	407,984	[0, 350]
Triage resprate (bpm)	20,356	404,725	[0, 300]
Triage O2sat (%)	20,637	404,438	[0, 100]
Triage SBP (mmHg)	18,307	406,774	[0, 375]
Triage DBP (mmHg)	19,486	405,595	[0, 375]
Triage pain	38,236	386,846	[0, 10]
Triage acuity	6,987	418,094	[1, 5]
ED temperature last (°C)	26,837	398,200	[26, 45]
ED heartrate last (bpm)	18,438	406,643	[0, 350]
ED resprate last (bpm)	18,923	406,158	[0, 300]
ED O2sat last (%)	29,102	395,979	[0, 100]
ED SBP last (mmHg)	18,687	406,403	[0, 375]
ED DBP last (mmHg)	18,936	406,145	[0, 375]
ED pain last	49,650	375,432	[0, 10]
ED LOS (h)	3,053	410,132	-

3.2.3 Outlier imputation experiment

Following the detection of outliers in both vital signs and ED Length of Stay (ED LOS) data, we conducted an extensive imputation experiment to benchmark various methods for handling missing values introduced by outlier detection. This experiment aimed to assess the accuracy of each imputation approach across these different types of clinical data.

3.2.3.1 Imputation methods evaluated

The following imputation methods were applied to handle the artificially induced missing values for both vital signs and ED LOS data:

- Median imputer: The median value for each variable was used to replace missing values. This approach is commonly used for its robustness against outliers.
- K-nearest neighbors (KNN) imputer: The KNN imputer utilized the nearest five neighbors (n_neighbors = 5) to impute missing values. This method allows for local patterns in the data to be leveraged when estimating the missing values (29).
- Iterative imputer: A multivariate approach that iteratively estimated missing values using relationships between all other features. This method is particularly suited for datasets with highly correlated features (29).
- Autoencoder imputer: A deep learning-based autoencoder (30) was implemented . The model architecture consisted of an input



10.3389/fdgth.2024.1498939

layer matching the number of features, a dense hidden layer with 10 nodes using ReLU activation, and a reconstruction layer. Training was carried out using the dataset with missing values filled by median imputation, with 50 epochs and a batch size of 32.

3.2.3.2 Evaluation metrics

To quantify the accuracy of each imputation method, two commonly used metrics were employed: Mean Squared Error (MSE) and Mean Absolute Error (MAE). These metrics compared the imputed values against the original values from the subset without missing data.

3.2.3.3 Experimental design

The experiment was carried out separately for both the vital signs and the ED LOS variables. In each case, a large subset (n = 50,000/250,000) of relevant data without missing values was selected for evaluation. To simulate missing data, 20% of the values in the subset were randomly set to missing using a probabilistic mask. Then the various imputation methods were used to predict values and compare them with the original known value. This process was repeated fives times to ensure robustness in the evaluation and average MSE and MAE were found.

3.3 Data set

A master dataset of 410,927 Emergency Department (ED) visits from 202,503 unique adult patients (aged over 18) was generated, which included 72 different variables. The resulting dataset was a tabular dataset with a shape of (410,927 \times 72), comprising both categorical and continuous features. Table 3 provides the baseline characteristics of the study population. The mean age of the patients in this dataset is 52.76 years, with a standard deviation of 20.61 years, indicating a broad age range ranging from 18 to 91 years. The racial makeup of our cohort offers crucial understanding of the study's demographics, with primary groups being 53.81% White, 18.21% Black or African American, and 4.87% classified as Other. The data set's gender breakdown shows a balanced representation, with a total count of 202,503 persons. 45.90% of these are men, while 54.10% are women. More information is provided in Table 3.

The outcome variable ED LOS was classified into two categories: short (less than 4.5 h) and long (greater than 4.5 h). This cutoff was chosen because it represents the average time hospitalized patients spend in the ED (22). According to relevant literature (31), the 4.5 h threshold aligns with several key performance metrics used in emergency care, which often consider a 4- to 5 h mark critical to evaluate quality of care and patient performance. The United Kingdom Department of Health (32) also supports the implementation of the 4 h rule.

TABLE 3 Baseline characteristics of the study population.

	n = 410,927
Demographics	
Age (years)	52.76 (20.61)
Gender, <i>n</i> (%)	
Female	222,318 (54.10%)
Male	188,699 (45.90%)
Charlson comorbidity index score	1.37 (2.63)
Elixhauser comorbidity index score	3.09 (4.79)
Healthcare utilization	
Number of ED visits in past 30 d	0.24 (0.78)
Number of ED visits in past 90 d	0.53 (1.61)
Number of ED visits in past 365 d	1.40 (4.21)
Number of Hospital visits in past 30 d	0.16 (0.52)
Number of Hospital visits in past 90 d	0.36 (1.03)
Number of Hospital visits in past 365 d	0.97 (2.70)
Number of ICU visits in past 30 d	0.02 (0.16)
Number of ICU visits in past 90 d	0.05 (0.27)
Number of ICU visits in past 365 d	0.11 (0.50)
Medication administered during the previous visi	t, n (%)
Yes	297,075 (72.29%)
No	113,852 (27.71%)
Triage vital signs	
Temperature (°C)	36.72 (0.54)
Heart rate (bpm)	85.22 (17.49)
Respiratory rate (bpm)	17.58 (2.50)
Oxygen saturation (%)	98.36 (2.40)
Systolic blood pressure (mmHg)	134.78 (22.13)
Diastolic blood pressure (mmHg)	77.39 (14.69)
Pain scale	4.11 (3.62)
Acuity	2.63 (0.71)
ED last vital signs	
Temperature (°C)	36.76 (0.37)
Heart rate (bpm)	78.19 (14.45)
Respiratory rate (bpm)	17.26 (2.48)
Oxygen saturation (%)	98.16 (2.93)
Systolic blood pressure (mmHg)	127.38 (19.56)
Diastolic blood pressure (mmHg)	73.55 (13.60)
Pain scale	2.18 (2.82)
Outcomes	
Length of stay (h), n (%)	
Short (≤ 4.5 h)	158,898 (38.67%)
Long (>4.5 h)	252,029 (61.33%)
Hospitalization, n (%)	
Yes	191,289 (46.55%)
No	219,638 (53.45%)
Critical, n (%)	
Yes	27,989 (6.81%)
No	382,938 (93.19%)
ED revisit 3 d, <i>n</i> (%)	
Yes	14,557 (3.54%)
No	396,370 (96.46%)

Continous variables presented as mean (sd) and categorical ones presented as count (percentage).

TABLE 4 Comparison of ED LOS dataset distribution before and after SMOTETomek application.

Outcome: ED LOS					
	Short (\leq 4.5 h)	Long (>4.5 h)			
Original data set (Im	balanced)				
Training data	127,168 (38.68%)	201,574 (61.32%)			
Test data	31,726 (38.60%)	50,459 (61.40%)			
Total	158,894 (38.67%)	252,033 (61.33%)			
Balanced data set (SMOTETomek)					
Training data	201,489 (50.00%)	201,489 (50.00%)			
Test data	31,726 (38.60%)	50,459 (61.40%)			
Total	233,315 (48.07%)	251,948 (51.93%)			

The outcome variable ED LOS had a mean value of 6.40, with a right-skewed distribution, as illustrated in Figure 2.

Initially, the data set was divided into training and test sets using a 80/20 ratio as detailed in Table 4. This method guarantees that the evaluation metrics accurately represent the model performance on unseen data.

3.4 Data balancing methods for ED LOS

In this study, the ED Length of Stay (ED LOS) was identified as an imbalanced outcome variable, with the majority of instances classified as long stays (62%). To address this imbalance, we applied several data balancing techniques using the imblearn library in Python (33), focusing solely on the outcome variable ED LOS.

3.4.1 Balancing methods

The following balancing methods were applied:

- SMOTE (Synthetic minority over-sampling technique): The Synthetic Minority Over-sampling Technique (SMOTE) (34) addresses imbalanced data by creating synthetic samples for the minority class based on nearest neighbors. This increases minority representation and contributes to a more balanced training set.
- ADASYN (Adaptive synthetic sampling): Adaptive Synthetic (ADASYN) (35) algorithm is similar to SMOTE, but it focuses on difficult-to-learn instances by creating synthetic samples in areas where minority classes are underrepresented.
- **Tomek links**: Tomek Links (36) is an undersampling method that eliminates overlapping samples from different classes, thereby improving class separation and lowering ambiguity.
- **SMOTETomek**: The SMOTE-Tomek method (12) combines SMOTE and Tomek Links to balance classes and remove overlapping samples, resulting in a better-defined dataset and improved classifier performance.
- SMOTEENN: The SMOTE-ENN method (37) combines SMOTE oversampling with Edited Nearest Neighbors (ENN) undersampling to generate synthetic minority samples and remove noisy or misclassified samples, improving data quality.

3.4.2 Evaluation of balancing techniques

We evaluated the balanced datasets using various machine learning models, including Random Forest, Gradient Boosting, Logistic Regression, and a Multi-Layer Perceptron (MLP) neural network. A 5-fold stratified cross-validation was used to ensure the reliability of the metrics. Evaluation metrics were calculated for both the training and the test sets.

3.4.3 Confusion matrix analysis

To better understand how data balancing affects model performance, we created a confusion matrix for each balancing technique and classifier. The confusion matrix provided a detailed view of true positives, true negatives, false positives, and false negatives, allowing us to evaluate the accuracy of each balancing strategy in correctly identifying both "Short" and "Long" ED LOS categories. Refer to Figure 3.

3.4.4 Experimental procedure

The balancing and evaluation procedure consisted of the following steps:

- 1. **Splitting the dataset:** The dataset was divided into training and testing sets in an 80/20 split. To provide an unbiased evaluation of model performance, the training set was balanced while the test set remained unchanged.
- 2. Cross-validation and resampling: During the 5-fold stratified cross-validation, each training fold was resampled using one of the balancing techniques. The balanced training set was then utilized to train the models and validated on the corresponding unbalanced test fold to calculate accuracy, AUC, specificity, sensitivity, and F1 score.

3.5 Feature engineering and selection

3.5.1 Feature engineering

In addition to the 72 analytical variables, several new variables were derived from the original data to improve the performance of the model. Notably, a new feature, *Medication Event*, was introduced as a boolean variable that indicates whether the patient received medication during a previous visit. The Charlson comorbidity index (CCI) and Elixhauser comorbidity index (ECI) were also calculated. For both indices, weights ranging from 1 to 6 were assigned to various comorbid conditions based on their severity (38). These weights were then summed together to generate a single score for each index, providing a comprehensive measure of each patient's comorbidity burden.

To facilitate prediction of the outcome, the length of stay ("ED_LOS") variable was divided into two categories. Specifically, a 4.5 h threshold was used to classify LOS as "short" (\leq 4.5 h) or "long" (>4.5 h). This classification produced a new binary variable in which short stays were encoded as 0 and long stays as 1, ensuring consistency for subsequent modeling, yielding 158,898 short stays (38.67%) and 252,029 long stays (61.33%).



3.5.2 Encoding categorical features

To allow for more effective modeling, categorical variables were encoded using one-hot encoding utilizing the pandas library (39). This transformation enabled categorical variables to be represented by binary columns, providing flexibility for models that benefit from binary feature representations. Features encoded include *Race, Arrival Transport, Disposition, Insurance,* and *Chief Complaint* indicators. Each category within these characteristics was represented as a separate binary column to ensure completeness and preservation of information. Table S1 of the Supplementary Material provides a complete summary of the one-hot encodings.

3.5.3 Feature selection

In order to perform feature selection, a comprehensive feature importance analysis was performed. This process involved taking advantage of the inherent ability of the Gradient Boosting (GB) model to rank features based on their predictive power (40). By analyzing the importance scores of the GB model, we identified the ten most influential variables for ED LOS. This step was critical in reducing the dimensionality, which simplified the data and increased the efficiency of subsequent analysis. By limiting the variables to those that were the most important, we not only improved the interpretability of our model, but also reduced computational complexity and the risk of over-fitting (41). Importance scores provided by the GB model highlighted key

TABLE 5	Description	of top	10 GB	importance	variables.
---------	-------------	--------	-------	------------	------------

Variable name	Description	Туре
Triage acuity	Triage acuity [emergency severity index (1: Resuscitation/2: Emergent/3: Urgent/4: Less urgent/5: Non-urgent)]	Categorical
Arrival transport unknown	Indicator for unknown mode of patient arrival to the ED (YES/NO).	Boolean
Elixhauser comorbidity index (ECI)	Elixhauser Comorbidity Index (ECI) score, representing the comorbidity burden of the patient.	Numeric
Disposition admitted	Indicator for whether the patient was admitted to the hospital from the ED (YES/ NO).	Boolean
Chief complaint abdominal pain	Indicator for whether the chief complaint at triage was abdominal pain (YES/NO).	Boolean
Arrival transport ambulance	Indicator for whether the patient arrived at the ED via ambulance (YES/NO).	Boolean
Age	Age of the patient.	Numeric
Medication event	Indicator for whether the patient was given medication in a prior visit (YES/NO).	Boolean
Disposition home	Indicator for whether the patient was discharged home from the ED (YES/NO).	Boolean
Disposition left without being seen	Indicator for whether the patient left the ED without being seen by a clinician (YES/NO).	Boolean

variables such as patient demographics, initial triage vital signs, and comorbidities that significantly affect ED LOS. The list of extracted variables and their description is provided in Table 5.

3.6 Machine learning and evaluation methods

3.6.1 Machine learning methods

Following the successful data preprocessing and feature selection, we evaluated four machine learning methods: Gradient Boosting (GB), Random Forest (RF), Logistic Regression (LR), and a Multi-Layer Perceptron (MLP) neural network. We used the scikit-learn package (42) for tree-based methods and logistic regression, and Keras (43) to implement the MLP model.

The MLP architecture, consisting of an input layer of 10 neurons and a single hidden layer of 64 neurons with ReLU activation as seen in Supplementary Figure S1, was inspired by previous work in healthcare prediction tasks (7). The hyperparameters and model parameters used for each machine learning method, and MLP, are summarized in Table 6.

3.6.2 Evaluation methods

To thoroughly assess the performance of our prediction models, we examined the receiver operating characteristic (ROC) curve and

TABLE 6 Machine learning modeling parameters.

Algorithm	Parameter	Default value
Logistic regression	Maximum iterations	1000
Random forest	No. of estimators	100
Gradient boosting	No. of estimators	100
Multi-layer perceptron	Learning rate	0.001
	Batch size	200
	Epochs	20

reported the area under the curve (AUC) as the primary indicator of overall predictive performance. Furthermore, additional evaluation metrics such as accuracy, sensitivity, specificity, and F1 score were calculated during model training and testing to gain a comprehensive understanding of each model's effectiveness.

In addition, to ensure robustness, we used five-fold stratified cross-validation during model training. This technique ensures that each fold has a similar class distribution, providing a more reliable evaluation of the model performance (44).

Furthermore, we used McNemar's test (45) to compare the performance of the models pairwise. McNemar's test, a non-parametric statistical test, was used to see if the differences in the misclassification rates between models were statistically significant. The contingency tables were created using the model predictions, and p-values less than 0.05 were used to indicate significant differences in model performance.

3.7 Rule extraction

The purpose of extracting rules from the ensemble techniques is to make the model decision logic clearer and more transparent to domain experts while retaining the predictive power and precision that the ensemble methods offer (46). In this paper, we used TE2Rules (47) rule extraction algorithm which has four primary stages:

- At first, TE2Rules generates initial rule candidates by associating rules with distinct TE tree nodes. These rules describe the decision paths that lead from the root to each node.
- The second stage assesses these candidates based on two criteria: the presence of positive instances covered by the rule and rule precision greater than a predefined threshold. Rules that do not meet these requirements are eliminated.
- In the third stage, TE2Rules develop stricter rule candidates by merging pairs of existing candidates. This process is repeated in phases, eventually refining the rules. Importantly, TE2Rules often converges to a high level of fidelity in a few stages, minimizing the requirement for in-depth processing.
- Finally, once TE2Rules has identified rules that cover all positive cases, it reduces the rule list by identifying the smallest subset of rules that cover all positives collectively. This phase minimizes redundancy and improves the interpretability of the rules, making it helpful for understanding the Tree Ensemble model's decision-making process (47).

3.8 Rules analysis

In order to substantiate the reliability and effectiveness of the rules derived from the Gradient Boosting model, we performed a comprehensive analysis of the rules by extracting and evaluating three key metrics. The generated metrics include overall coverage, relative coverage, and rule accuracy, each of which is crucial to evaluate the practical utility and precision of the rules.

3.8.1 Overall coverage

Overall coverage quantifies the proportion of the data set that is covered by a given rule (10). It indicates how extensively a rule applies across the entire dataset. A higher coverage suggests that the rule is applicable to a larger segment of the data, which can be indicative of its general relevance.

$$Overall \ coverage = \left(\frac{Number \ of \ samples \ where \ rule \ is \ activated}{Total \ number \ of \ samples}\right) \times 100$$

3.8.2 Relative coverage

Relative coverage measures the proportion of samples in a specific category that are covered by the rule (48). This metric helps us understand how well the rule performs for a particular category such as long or short ED stays, reflecting the rule's effectiveness in distinguishing between different categories within the dataset.



3.8.3 Rule accuracy

Rule Accuracy measures the proportion of correctly classified samples out of all samples where the rule is activated (10). This metric evaluates how accurately the rule identifies the intended category and misclassified. Greater accuracy demonstrates the rule's effectiveness in producing correct predictions upon activation.

Rule accuracy =
$$\left(\frac{\text{Number of correctly classified samples}}{\text{Number of activated samples}}\right) \times 100$$

4 Results

In this section, we present the findings of our investigations. We begin by presenting our experimental results, feature analysis to gain a better understanding of the data set. Next, we showcase the benchmark task's models and derived rules.

4.1 Imputation experiment results

To evaluate the effectiveness of different imputation methods, we performed an extensive imputation experiment. Table 7 summarizes the average results from the experiments. Notably, the Iterative Imputer performed the best, with the lowest average MSE and MAE across both sample sizes (50,000 and 250,000) and was used for replacing missing values earlier detected. The KNN Imputer and Median Imputer performed similarly but slightly underperformed compared to the Iterative Imputer. Despite its complexity, the Autoencoder Imputer had a higher

TABLE 7 Average performance of imputation methods.

Sample size	Imputation method	Average MSE	Average MAE
50,000	Median imputer	24.89	1.23
	KNN imputer	26.38	1.27
	Iterative imputer	16.45	0.98
	Autoencoder imputer	32.10	2.79
250,000	Median imputer	24.83	1.23
	KNN imputer	27.23	1.29
	Iterative imputer	16.44	0.98
	Autoencoder imputer	26.68	1.85

Boldface denotes lowest average MSE/MAE.

error rate, indicating that its performance may not be as generalizable for this dataset.

4.2 Balancing datasets experiment results

The evaluation of various sampling methods combined with machine learning models to predict ED LOS yielded insightful findings. The sampling techniques employed were ADASYN, SMOTE, SMOTEENN, SMOTETomek, and TomekLinks. The performance metrics for testing and training–AUC, Accuracy, F1 Score, Sensitivity, and Specificity–are detailed in Supplementary Materials S2, S3, with confusion matrices available in Figure 3

SMOTETomek emerged as the most balanced method across all metrics. SMOTETomek consistently achieved high Area Under the Curve (AUC) scores across models, with the MLP model having the highest AUC of 0.731, followed by Gradient Boosting at 0.729. SMOTETomek's specificity was also consistently high across multiple models, achieving 65.65% with LR, demonstrating its ability to accurately identify patients predicted to have a short ED LOS. The confusion matrix for SMOTETomek combined with MLP also advocated for its balanced performance, with 64.38% of short LOS and 70.53% of long LOS correctly predicted, indicating that SMOTETomek could maintain a well-distributed prediction across both classes while maintaining relatively low FPs and FNs.

The analysis of the confusion matrices (Figure 3) provides additional support for these findings. SMOTETomek combined with GB produced a well-balanced prediction result, with 59.10% of the short LOS cases correctly identified and 74.86% of the long LOS cases correctly predicted. Similarly, SMOTETomek with LR correctly identified 64.15% of long LOS cases and predicted short LOS with 65.39% accuracy, achieving higher specificity than sensitivity in this case. Another notable result was TomekLinks combined with GB, which correctly predicted 88.78% of long LOS cases, the highest among all models and sampling methods, demonstrating its efficacy in detecting patients who may require prolonged hospital stays. TomekLinks + MLP also had a high true positive rate of 87.97%, demonstrating its utility in accurately detecting long stays. In contrast, ADASYN outperformed all other classifiers in terms of specificity. Specifically, ADASYN combined with Logistic

Regression achieved the highest specificity of 66.68% among all methods and classifiers, while ADASYN combined with Gradient Boosting demonstrated a strong AUC of 0.6998 and a specificity of 60.61%. However, this came at the expense of sensitivity, which was consistently lower than in other sampling methods.

Logistic regression, when combined with SMOTE, ADASYN, and SMOTETomek, consistently achieves greater specificity than sensitivity, as shown in Table S2. This distinct pattern, which does not appear with other classifiers, suggests that LR is effective in distinguishing Short ED LOS while maintaining a reasonably balanced recall for Long ED LOS. For example, SMOTE with LR correctly predicts 65.84% of short ED LOS cases compared to 63.60% to long ED LOS. Specificity outperformed sensitivity for the mentioned sampling methods, indicating that the LR classifier was better at correctly predicting Short ED LOS than Long ED LOS.

Overall, SMOTETomek appears to provide the most balanced performance, with high AUC, Sensitivity, and relatively high Specificity across multiple classifiers. Although TomekLinks had the highest sensitivity, it tended to sacrifice Specificity, making SMOTETomek the more robust option when considering overall balance. Other methods, such as SMOTEENN and ADSYN, also performed well, particularly in maintaining a high F1 score, which may be appropriate when balancing precision and recall is the main goal. This balancing experiment ensured that the models developed to predict ED LOS could learn equally from both categories, resulting in better and more actionable clinical insights.

4.3 Feature analysis

Our analysis provides critical information on the factors that influence ED LOS. It contains detailed information about patient characteristics, including a broad age range and various influencing variables. Table 8 shows key variables related to ED LOS classified as short (\leq 4.5 h) and long (>4.5 h), along with their Gradient Boosting (GB) importance scores and *t* statistics where continuous variables are presented as mean (sd), while categorical variables are presented as count (percentage). All variables were found to be statistically significant with a p-value <0.01, as indicated by t statistics, underscoring their influence on ED LOS.

The t statistic indicates significant differences between the short and long stay groups in all variables. The mean triage acuity scores are slightly higher for patients with short stays (2.78) than long

TABLE 8 Summary statistics and gradient boosting (GB) importance for ED LOS outcomes.

Variable name	Outc	ome	T-Statistic GB importance	
	ED LOS			
	Short (≤4.5 h)	Long (>4.5 h)		
Triage acuity	2.78 (0.79)	2.53 (0.63)	109.311 (S)	0.397
Disposition admitted			-116.812 (S)	0.118
YES	42,837 (26.96%)	112,957 (44.82%)		
NO	116,061 (73.04%)	139,072 (55.18%)		
Elixhauser comorbidity index score (ECI)	2.10 (4.06)	3.73 (5.10)	-107.314 (S)	0.102
Arrival transport unknown			90.953 (S)	0.101
YES	18,316 (6.49%)	3,334 (1.32%)		
NO	148,582 (93.51%)	248,695 (98.68%)		
Chief complaint abdominal pain			-81.656 (S)	0.077
YES	10,212 (6.43%)	37,063 (14.71%)		
NO	148,686 (93.57%)	214,966 (85.29%)		
Arrival by ambulance			-98.076 (S)	0.061
YES	43,313 (27.26%)	106,363 (42.20%)		
NO	115,585 (72.74%)	145,666 (57.80%)		
Age	48.19 (20.47)	55.65 (20.17)	-114.811 (S)	0.053
Medication administered during previous vis	it		-105.107 (S)	0.040
YES	100,382 (63.17%)	199,693 (78.04%)		
NO	58,516 (36.83%)	55,336 (21.96%)		
Disposition left without being seen			83.000 (S)	0.033
YES	5,092 (3.20%)	427 (0.17%)		
NO	153,806 (96.80%)	251,602 (99.83%)		
Disposition home			94.958 (S)	0.018
YES	104,881 (66.01%)	128,794 (51.10%)		
NO	54,017 (33.99%)	123,325 (48.90%)		

All variables were found to be statistically different (S) with p-value < 0.01.

stays (2.53), with a high GB importance score of 0.397. 6.49% of the patients with a short stay arrived by unknown means of transport, compared to 1.32% of the long-term patients, with an importance score of 0.083. The Elixhauser Comorbidity Index Score averages 2.10 for shorter stays and 3.75 for longer stays, with an importance score of 0.102. Furthermore, 44.82% of the long-term patients were later admitted, compared to 26.96% of the patients with a short-term stay, with an importance score of 0.118. Moreover, 27.26% of patients who had a short stay arrived by ambulance, whereas 42.20% of long stay patients arrived by ambulance, with an importance score of 0.061.

Additionally, 63.17% of patients who had a short stay in the ED were recorded to have received medication during a previous visit, compared to 78.04% of those with a long stay, with an importance score of 0.040. Moreover, 66.01% of the patients with a short stay were discharged home, while 51.10% of patients with a long stay experienced the same result, whereas 3.20% of patients with short stays left without being seen, compared to 0.17% of patients with long stays. The importance scores for these are 0.018 and 0.033, respectively.

4.4 Model evaluation

The top 10 previously identified variables were used to categorize ED LOS. Table 4 shows the distribution of training and test data based on ED LOS outcomes. In the original unbalanced data set, the training data have 61.32% long stays and 38.68% short stays, while the test data have 61.40% long stays and 38.60% short stays. After using SMOTETomek, the training set has a perfectly balanced distribution of 50.00% long

stays and 50.00% short stays. In general, the data set had a distribution of 61.33% long stays and 38.67% short stays in the original data, vs. 51.93% long stays and 48.07% short stays in the balanced data set of SMOTETomek.

Next, we evaluated various machine learning models for predicting ED LOS using both the original unbalanced data set and a balanced version generated by SMOTETomek. The evaluation uses metrics such as accuracy, sensitivity, specificity, AUC score, and F1 score, as shown in Table 9.

GB performed the best on the unbalanced dataset, with an AUC score of 0.730, F1 score of 0.782, accuracy of 69.93%, and sensitivity of 88.20%. This indicates that GB had strong predictive power, particularly in distinguishing between patients with long and short ED stays. However, the specificity for GB remained low at 40.95%, indicating a tendency to overestimate the majority class (long stays) at the expense of accurately predicting the minority class (short stays). RF also performed well, with an AUC of 0.701 and the highest specificity (50.11%) among the models, implying that it was more effective in correctly identifying patients with short stays than GB. MLP also did well obtaining the highest accuracy on the original dataset (69.71%). MLP performed similar to LR in terms of Specificity, but MLP slightly outperformed LR.

The use of SMOTETomek significantly altered the model dynamics. GB continued to outperform other models on the balanced dataset, with the highest accuracy (68.86%), sensitivity (75.36%), and F1 score (0.748). This suggests that GB was able to maintain its robustness despite the use of SMOTETomek. RF also performed well in the balanced context, achieving one of the highest specificity (63.55%) but with a notable trade-off in sensitivity. LR model improved its specificity to 65.65%, the

TABLE 9 Average of 5-folds model performance on original and balanced datasets.

Model	Accuracy	Sensitivity	Specificity	AUC score	F1 score			
Original data set (ave	Original data set (average training scores \pm S.D)							
Gradient boosting	$69.98\% \pm 0.0006$	$88.24\% \pm 0.005$	$41.01\% \pm 0.009$	0.731 ± 0.0007	0.783 ± 0.0006			
Random forest	73.29% ± 0.03	$83.67\% \pm 0.002$	56.82% ± 0.003	0.805 ± 0.0004	0.794 ± 0.0005			
Logistic regression	$69.30\% \pm 0.0002$	$85.81\%~\pm~0.0008$	$43.12\% \pm 0.001$	0.699 ± 0.0002	$0.774~\pm~0.0002$			
MLP	$69.75\% \pm 0.002$	83.59% ± 0.029	47.80% ± 0.042	0.730 ± 0.001	0.772 ± 0.007			
Original data set (ave	erage test scores \pm S.D)							
Gradient boosting	69.93% ± 0.001	88.20% ± 0.005	$40.95\% \pm 0.008$	0.730 ± 0.002	0.782 ± 0.001			
Random forest	67.99% ± 0.002	79.26% ± 0.002	50.11% ± 0.007	0.701 ± 0.0016	0.752 ± 0.0008			
Logistic regression	69.29% ± 0.0008	$85.81\% \pm 0.0009$	43.10% ± 0.003	0.699 ± 0.001	0.774 ± 0.0004			
MLP	69.71% ± 0.003	$83.51\% \pm 0.029$	$47.82\% \pm 0.040$	0.730 ± 0.002	$0.772~\pm~0.008$			
Balanced data set (Sl	MOTETomek) (average trai	ning scores \pm S.D)						
Gradient boosting	$68.89\% \pm 0.001$	75.39% ± 0.004	$58.59\% \pm 0.004$	0.730 ± 0.0006	$0.748~\pm~0.002$			
Random forest	$71.54\% \pm 0.0005$	$71.58\% \pm 0.001$	$71.48\% \pm 0.001$	0.800 ± 0.002	0.755 ± 0.0005			
Logistic regression	$64.61\% \pm 0.002$	$63.99\% \pm 0.005$	$65.59\% \pm 0.004$	$0.699~\pm~0.0001$	0.689 ± 0.003			
MLP	$68.55\% \pm 0.002$	$72.89\% \pm 0.013$	$61.66\% \pm 0.015$	$0.732~\pm~0.0011$	$0.740~\pm~0.005$			
Balanced data set (SMOTETomek) (average test scores \pm S.D)								
Gradient boosting	$\mathbf{68.86\%} \hspace{0.2cm} \pm \hspace{0.2cm} 0.002$	75.36% \pm 0.002	$58.55\% \pm 0.003$	$0.729~\pm~0.002$	$\textbf{0.748}~\pm~\textbf{0.002}$			
Random forest	65.43% ± 0.0003	66.61% ± 0.0014	63.55% ± 0.002	0.700 ± 0.002	0.703 ± 0.0005			
Logistic regression	$64.64\% \pm 0.001$	$64.00\% \pm 0.004$	$\mathbf{65.65\%}~\pm~0.004$	0.699 ± 0.001	0.689 ± 0.003			
MLP	68.53% ± 0.003	72.90% ± 0.013	61.60% ± 0.013	$\textbf{0.731}~\pm~\textbf{0.001}$	$0.740~\pm~0.005$			

Boldface denotes best performing model with respect to metric.

10.3389/fdgth.2024.1498939

highest among all models. However, sensitivity decreased, indicating an increased focus on correctly predicting short stays at the expense of long stays.

Training scores across models followed consistent trends with test scores, demonstrating the reliability of the evaluation metrics. For example, MLP had the joint highest training AUC (0.730) in the unbalanced dataset, which corresponded to its strong performance in specific metrics during testing. Interestingly, the F1 scores for GB and MLP remained relatively high in both the original and balanced datasets, demonstrating the models' consistent ability to balance precision and recall effectively.

McNemar's test was run on both the unbalanced and balanced datasets results to better understand the differences in model performance. The results, which are summarized in **Supplementary Table S4**, show that for the majority of comparisons, the differences in model performance were statistically significant, with the exception of GB vs MLP on the balanced dataset, where the difference was not significant. This implies that GB and MLP performed similarly on the balanced data, indicating that both models were equally capable of capturing the nuances of ED LOS predictions and the difference in performance was not statistically significant.

In addition, we conducted minority class analysis for the short stay. Although the labels were flipped and minority class was now "long stay," the overall model performance has remained largely consistent, demonstrating robustness to change. The specificity had generally increased in comparison to the sensitivity, indicating a reversal of focus, with the models now correctly predicting the minority class (long stays) more frequently, at the expense of sensitivity. For example, Gradient Boosting (GB) had a specificity of 88.43% and a sensitivity of 40.76% for the minority class, indicating that it was more accurate in predicting "long stay" cases. Similarly, Random Forest (RF) showed 79.75% specificity and 49.22% sensitivity for the minority class. This is a significant but expected contrast to the previous evaluation, in which sensitivity was higher, indicating a shift in the models' prediction tendencies. Furthermore, Logistic Regression (LR) and Multi-Layer Perceptron (MLP) demonstrated similar trade-offs, with both models having higher specificity than sensitivity, highlighting the consistent trend summarized in Supplementary Table S5.

4.5 Rules extracted

We used TE2Rules (47) to derive a global rule list from a 100-tree Gradient Boost Model, with a precision threshold of 90% set. Table 10 provides a detailed summary of the rules and fidelity metrics for predicting Emergency Department Length of Stay (ED LOS) for both long and short outcomes. Six rules for long stays explain 88.16% of the overall predictions, including all positive predictions and 48.30% of the negative predictions. The key variables in these rules include age, type of arrival transport, triage acuity, and chief complaint.

In contrast, the seven rules for short stays explain 96.91% of the overall predictions. These rules cover 93.99% positive predictions and 97.29% negative predictions. Factors such as arrival

TABLE 10 Rules and metrics for short (\leq 4.5) and Long (>4.5 h) Outcomes. (see Table 12 for rule performance metrics).

Outcome	 Details
Short stay	Number of rules: 7
(ED LOS \leq 4.5 h)	Overall prediction explanation: 96.91%
	Positive prediction explanation: 93.99%
	Negative prediction explanation: 97.29%
Rules	Description
Short rule 1	Arrival transport ambulance = NO & chief complaint abdominal pain = NO
	& Elixhauser comorbidity index (ECI) score ≤ 0.5 & medication event = NO
	& Triage acuity = [urgent/less urgent/non-urgent]
Short rule 2	Age \leq 30.5 & arrival transport ambulance = NO
	& Chief complaint abdominal pain = NO & triage acuity = [urgent/less urgent/non-urgent]
Short rule 3	Triage acuity = [less urgent/non-urgent]
Short rule 4	Arrival transport unknown = YES
Short rule 5	$32.5 < Age \le 37.5$ & Arrival transport ambulance = NO
	& Chief complaint abdominal pain = NO & elixhauser
	comorbidity index (ECI) score ≤ 0.5
	& Triage acuity = [urgent/less urgent/non-urgent]
Short rule 6	$38.5 < Age \le 45.5$ & arrival transport ambulance = NO
	& Arrival transport unknown = NO & chief complaint abdominal pain = NO
	& Disposition admitted = NO & elixhauser comorbidity index (FCI) score < 0.5
	& Triage acuity = [urgent]
Short rule 7	Disposition left without being seen = YES
Long stay	Number of rules: 6
(ED LOS > 4.5 h)	Overall prediction explanation: 88.16%
	Positive prediction explanation: 100%
	Negative prediction explanation: 48.30%
Rules	Description
Long rule 1	Age > 29.5 & Arrival transport unknown = NO
	& Triage acuity = [resuscitation/emergent/urgent]
Long rule 2	Elixhauser comorbidity index (ECI) score > 0.5
Long rule 3	Arrival transport ambulance = YES
Long rule 4	Chief complaint abdominal pain = NO
	& Disposition home = NO & Disposition left without being seen = NO
Long rule 5	Chief complaint abdominal pain = YES
Long rule 6	Age \leq 70.5 & Triage acuity = [resuscitation/emergent]

TABLE 11 Summary of rule activation and stay categories.

Category	Short stay (≤4.5 h)	Long stay (>4.5 h)
No rules activated (short stay)	314,341	-
No rules activated (long stay)	-	47,222
No rules activated (both)	0	0
Dilemma points (both rules activated)	49,364	49,364

transport type, chief complaint of abdominal pain, and triage acuity also play significant roles in these rules. Each rule is detailed with specific conditions, allowing a clear view of the model's decision boundaries for distinguishing between short and long ED LOS outcomes.

Furthermore, Table 11 summarizes the data observations related to the activation of a certain rule. Notably, there are no

Rule	Overall coverage (%)	Category	Relative coverage (%)	Accuracy (%)
Short rule 1	10.54	Short	7.30	69.27
		Long	3.24	30.73
Short rule 2	10.10	Short	6.77	66.97
		Long	3.34	33.03
Short rule 3	7.06	Short	5.74	81.30
		Long	1.32	18.70
Short rule 4	3.32	Short	2.51	75.58
		Long	0.81	24.42
Short rule 5	2.28	Short	1.42	62.23
		Long	0.86	37.77
Short rule 6	1.75	Short	0.97	55.33
		Long	0.78	44.67
Short rule 7	1.34	Short	1.24	92.26
		Long	0.10	7.74
Long rule 1	74.78	Short	23.49	31.42
		Long	51.28	68.58
Long rule 2	43.24	Short	12.43	28.75
		Long	30.81	71.25
Long rule 3	36.42	Short	10.54	28.94
		Long	25.88	71.06
Long rule 4	18.11	Short	4.61	25.46
		Long	13.50	74.54
Long rule 5	11.50	Short	2.49	21.60
		Long	9.02	78.40
Long rule 6	7.45	Short	2.77	37.21
		Long	4.68	62.79

TABLE 12 Rules empirical metrics (see Table 10 for rules).

cases in which rules for both short and long stays are not activated, and a total of 49,364 observations show simultaneous activations for both short and long rules, indicating dilemma points.

Moreover, Table 12 provides metrics for each rule, including coverage and accuracy. For long stay rules, rule 5 has the highest overall accuracy at 78.40% whereas long rule 1 has the highest overall coverage where it covers 74.78% of the data. Conversely, for short stay rules, rule 7 has the highest accuracy at 92.26%, despite having a lower overall coverage of 1.34%. This contrast highlights the varying efficacy of rules in predicting different outcomes. long stay rules have higher coverage but vary significantly in accuracy, with some rules achieving high coverage while having lower accuracy. Short stay rules, on the other hand, can achieve very high accuracy despite covering less of the dataset, indicating that they are more specialized in identifying true short stay cases.

5 Discussion

5.1 Summary of key findings

This section presents the key findings of the research extensively. we start by examining the most important variables of our analysis, moving towards model evaluation and finally rules analysis.

5.1.1 Key variables influencing ED LOS

The study identified several critical variables that significantly influence the LOS of the ED, with the following the most impactful:

- 1. Triage acuity: Patients with higher acuity levels were found to have shorter LOS, likely due to prioritization in treatment and faster resolution of less serious conditions.
- 2. Elixhauser comorbidity index (ECI): Significantly higher ECI scores correlated with longer LOS, reflecting the increased complexity and resource demands associated with treating patients with multiple comorbidities.
- 3. Arrival methods: Patients who arrived via ambulance or other known transport means had longer LOS, indicative of the more severe nature of their conditions requiring extended medical attention. Whereas, patients arriving with unknown means of transport had a shorter stay in the ED.
- 4. Previous medication history: A history of medication during previous ED visits was a strong predictor of longer LOS, indicating that ongoing treatment needs significantly influences the duration of stay.

5.1.2 Model performance and comparison

5.1.2.1 Gradient boosting (GB) and Multi-layer perceptron (MLP)

Gradient Boosting (GB) and Multi-Layer Perceptron (MLP) outperformed all other models tested, with high AUC scores, sensitivity, and consistent reliability across multiple metrics. GB performed slightly better overall, with an AUC score of 0.730, accuracy of 69.93%, sensitivity of 88.20%, and specificity of 40.95%. This suggests that GB was particularly effective at correctly predicting long-stay cases, while its specificity was relatively higher, indicating a greater emphasis on avoiding false positives, which is critical in emergency department settings. TomekLinks sampling paired with GB achieved the highest test sensitivity of 75.36%, making this model combination particularly suitable for scenarios where identifying all patients needing long stays is critical. However, this comes at the expense of specificity, which leads to a higher number of false positives for short stays.

Similarly, MLP showcased competitive performance with an AUC of 0.730, accuracy of 69.71%, and sensitivity of 83.51% and specificity of 47.82%. Although it did not surpass GB, MLP's ability to balance between sensitivity and specificity makes it a strong candidate for ED LOS classification. Confusion matrix analysis for SMOTETomek + MLP reveals a balanced trade-off between sensitivity and specificity, indicating that it can correctly classify both long and short stays. For example, 72.90% of long stays were correctly predicted, and the true negative rate was also relatively high, resulting in fewer unnecessary prolonged hospitalizations than TomekLinks.

5.1.2.2 Random forest (RF)

The Random Forest (RF) model performed well, but fell short of GB and MLP in terms of overall metrics. With an AUC of 0.701 and a specificity of 50.11%, RF was more effective in identifying patients who needed a short stay, providing a relatively balanced trade-off between true positives and true negatives. However, the sensitivity of 79.26% remained relatively moderate, indicating a slight weakness in accurately predicting long stays compared to the other models. The confusion matrices show that RF benefited

significantly from the use of SMOTETomek, resulting in a balanced classification for both the ED and LOS classes. The true negative rate was 63.55%, indicating improved specificity after resampling.

According to Bentéjac et al. (49), gradient boosting outperforms random forest in terms of generalization, especially when dealing with class imbalances. GB's sequential nature allows for iterative error correction, which makes it better suited to the task of discriminating between short and extended ED stays (13, 17). This is consistent with our findings, in which GB outperformed RF in terms of accuracy and AUC. Gradient boosting has the advantage due to its focus on misclassified samples during training, allowing for more efficient adaptation to the minority class (49).

5.1.2.3 Logistic regression (LR)

Logistic Regression (LR) served as a baseline model and showed high specificity, especially when combined with SMOTETomek and ADASYN. LR had a test accuracy of 69.71% and a sensitivity of 85.81% on the unbalanced dataset, indicating its ability to accurately predict long stays and identify high-risk patients. However, the model lacked specificity (43.10%), making it unsuitable for predicting all patients who may require short stays.

LR consistently demonstrated higher specificity than sensitivity across various sampling methods, which corresponds to its lower complexity and linear decision boundary, effectively distinguishing short stays but missing long stays. The confusion matrices show that LR's performance on the sampled data is characterized by a higher true negative rate, demonstrating its ability to prioritize minimizing false positives. The ADASYN + LR combination had the highest specificity of 66.68%, highlighting its utility in situations where reducing unnecessary admissions(short stays) is a priority.

5.1.3 Confusion matrix analysis and practical implications

The analysis of confusion matrices reveals the various tradeoffs each model makes between sensitivity and specificity. SMOTETomek emerges as a robust balancing method, especially when combined with MLP and GB, ensuring that long stays are properly identified while not jeopardizing short stay detection. This makes SMOTETomek suitable for clinical applications in which it is critical to identify high-risk patients while maintaining resource efficiency.

TomekLinks, when combined with GB or MLP, produced high sensitivity rates, indicating that it is effective in identifying the majority of patients who require extended care. However, this resulted in an increased false positive rate, implying that it should be used when patient safety takes precedence over resource allocation. In contrast, ADASYN combined with LR or RF provided greater specificity, emphasizing its efficacy in situations where reducing unnecessary long stays is critical to managing emergency department capacity.

Finally, GB and MLP consistently outperformed in terms of balanced performance, with both demonstrating balanced superior metrics. The clinical goal should determine the model and sampling technique; for example, if the goal is to reduce missed high-risk cases, TomekLinks with GB or MLP would be ideal. On the other hand, SMOTETomek combined with GB or MLP remains the best option for balanced, general applicability, ensuring a moderate trade-off between sensitivity and specificity.

5.1.4 Rule extraction and analysis 5.1.4.1 Overview of rules

A key innovation in this study was the use of a rule extraction technique on the Gradient Boosting model, which was intended to turn complex model decisions into human-interpretable rules. The TE2Rules algorithm (47) extracted 13 rules (6 long stay and 7 short stay) that encapsulate significant patterns and relationships identified by the model. The rules were developed using key predictors such as triage acuity, age, arrival methods, and specific patient conditions, and were intended to provide actionable insights for emergency department decision-making.

The extracted rules serve two primary functions: improving model interpretability and increasing transparency in clinical decision-making. By translating a Gradient Boosting model's decision boundaries into a set of simple rules, healthcare professionals can gain a better understanding of why certain decisions were made. For example, these rules can be used to determine which patients are likely to spend long or short periods of time in the emergency department (ED), assisting with resource allocation and prioritizing patient care.

The rules were evaluated using several key metrics to determine their utility and reliability, including Overall Coverage. Higher overall coverage indicates that a rule is widely applicable across the dataset, implying general relevance. Rule 1 (Long Stay) had 74.78% coverage, indicating it applies to a significant portion of patient data. The broad applicability of the rule makes it especially useful for general triage decision-making. In addition, the accuracy of the rule is defined as the percentage of correctly classified samples among all samples where the rule is activated. High rule accuracy indicates that the rule is very effective in predicting the correct outcome in those cases. For instance, Rule 7 (Short Stay) had an extremely high rule accuracy of 92.26%, underscoring its strong predictive power in the subset it covers.

5.1.4.2 Extracted rules and their implications

The extracted rules revealed important patterns, for instance:

- Mid aged or older patients (>29.5) with critical triage acuity ratings [Resuscitation/Emergent/Urgent] and known arrival methods are likely to experience long stays (Rule 1, Long Stay). Such rules allow ED staff to preemptively allocate more resources to patients fitting these criteria.
- Conversely, specific scenarios like patients leaving without being seen, strongly correlated with shorter stays (Rule 7, Short Stay). This insight can be used to identify potential gaps in care, such as situations in which patients may leave due to excessive wait times.

The rules have significant practical implications, particularly in terms of increasing transparency and trust in clinical machine learning models (8, 9). The model supports and enhances clinical judgment by providing healthcare professionals with easily

interpretable rules. These rules could help with triage decisions, predict resource demands, and ultimately, improve patient outcomes by allowing for targeted interventions.

5.2 Comparison with previous studies

In this section, we compare the results of our study with those of previous research in predicting ED LOS. This analysis highlights the contributions of our methodology, especially in terms of the performance, interpretability, and importance of the identified factors. The studies are summarized in Table 1.

5.2.1 Overview of methodologies and models

Several studies have utilized machine learning approaches to predict ED LOS, employing a spectrum of models ranging from conventional statistical techniques to advanced deep learning frameworks. For instance, Pedro et al. (15) used a PRO-AGE scoring system combined with bootstrapping to predict LOS in older adults, achieving an AUC of 0.79. Ala et al. (13) focused their efforts on fast-tracking patients (<4 h) in the ED, using models such as MLP, CART, and SGB which achieved a an AUC of 0.752. In contrast, Zebin et al. (14) employed an auto-encoded deep neural network, achieving an accuracy of 77.7%.

In our study, similar to the work of Etu et al. (17), we applied GB models, which consistently surpassed other models in terms of Accuracy and AUC scores, demonstrating their robustness in predicting ED LOS. Additionally, unlike Rahman et al. (19), who used a decision tree model to predict ED LOS greater than 4 h with an accuracy of 84.94%, our approach also integrates various approaches to handle class imbalance and derives transparent rule metrics, improving generalizability and transparency of our models.

5.2.2 Feature selection and importance

The variables used in previous studies vary, but often include vital signs, demographic information, and admission details. For example, Ala et al. (13) highlighted chief complaints, age, and vital signs as crucial predictors, which aligns with the features we selected. Similarly, Zebin et al. (14) and Bopche et al. (16) emphasized the importance of demographics and ICD-9 diagnosis codes (chronic diseases), which are also consistent with our findings. Furthermore, factors such as age, mode of transport and triage acuity were identified as key predictors, supporting the findings of other studies, such as Chang et al. (18), who also highlighted similar variables as important in the prediction of ED LOS.

5.2.3 Comparison of model performance

Gradient Boosting (GB) and Deep Learning models are frequently reported as top performers (7, 13, 14, 16, 17, 20). As an illustration, Etu et al. (17). reported a GB model with an AUC of 0.93, outperforming other tree-based classifiers such as Decision Trees and Random Forest in both imbalanced and SMOTE-balanced datasets. Similarly, our study demonstrated that the GB models provided the highest accuracy and AUC scores, reinforcing their effectiveness in ED LOS classification.

However, unlike previous studies, our research goes beyond typical performance metrics by integrating a novel rule extraction process (46). This approach improves interpretability and provides transparent actionable insights for clinical decision making (11). This process translates the complex decision logic of the model into simple, human-interpretable rules. Each extracted rule was analyzed for its overall coverage, relative coverage, and rule accuracy, metrics that have not been previously implemented in the context of ED LOS prediction. Few studies, such as Rahman et al. (19) and Bopche et al. (16), have explored this avenue using the J48 decision tree algorithm and SHAP, respectively. Our work builds upon this by applying and evaluating human-readable rules that require minimal expert interpretation, using various metrics for comprehensive assessment.

5.2.4 Comparison of explainability methods

In the field of ED LOS prediction, explainability methods such as SHAP (Shapley Additive Explanations) and LIME (Local Interpretable Model-agnostic Explanations) have been used to clarify machine learning model decisions (16, 50, 51). For example, SHAP has been used to interpret feature contributions in predicting in-hospital mortality, long-term LOS and sepsis onset, providing consistent explanations across models. Similarly, LIME has been used to provide local approximations of model behavior, aiding understanding of individual predictions.

Our study distinguishes itself by incorporating the TE2Rules algorithm for rule extraction from a Gradient Boosting model. TE2Rules, in contrast to SHAP and LIME, which focus on feature importance and local approximations, translates complex model decisions into human-readable rules (46). This approach not only enhances interpretability but also enables actionable decision-making through explicit decision rules that clinicians can directly apply (52, 53).

For instance, TE2Rules generates simple "if-then" rules that summarize the underlying patterns in the data, making them easy to understand for healthcare professionals. Compared to feature importance scores (SHAP) or localized surrogates (LIME), TE2Rules offers a distinct advantage in transforming black-box models into user friendly rule-based logic. This rulebased explanation can directly guide treatment strategies or triage decisions, eliminating the need for additional interpretation layers. Increased transparency also helps with clinical accountability, as decision rules can be validated against medical guidelines and expert judgment, fostering trust in AIdriven recommendations.

5.2.5 Novel contributions

Our work provides a thoughtful combination of methodological advancements and practical innovations that set it apart from previous research in the field. We offer several novel contributions that primarily aim to improve the interpretability, robustness, and practical applicability of predictive models for ED LOS:

10.3389/fdgth.2024.1498939

- Explainable AI techniques: We used the TE2Rules technique for rule extraction, which greatly improves the interpretability of ensemble methods such as Gradient Boosting. This technique improves healthcare practitioner's understanding of the decision-making process and builds trust in the model. Unlike traditional approaches that focus solely on model performance metrics like AUC, sensitivity, and specificity, our analysis offers a more granular analysis by dissecting the performance of each individual rule. This not only improves the interpretability of the model, but also provides a practical framework for applying these rules in real-world scenarios (10, 48). To our knowledge, the use of TE2Rules in predicting ED LOS is novel, as previous research has primarily focused on other explainability methods (50). For instance, some studies have used SHAP to interpret models that predict ED LOS, emphasizing feature importance but lack the simplicity of rule-based explanations.
- Use of limited, clinically relevant features: Unlike many previous studies, which relied on a relatively large number of variables (15–19), we used only 10 readily available and clinically significant features. This not only improves model simplicity, but also makes it easier to implement in real-world clinical settings while maintaining model accuracy and utility.
- Robust evaluation on balanced and imbalanced data: We evaluated model performance on both balanced and real-world imbalanced datasets in an exhaustive manner to ensure the findings' robustness and practical relevance.

5.3 Limitations and future directions

Although the study demonstrated the effectiveness of rule extraction and model interpretability, it is necessary to highlight several limitations and suggest future avenues.

This study relies on historical data from a single source, which could introduce biases into model predictions and restrict the generalizability of the results. The performance of the different methods used in this study may vary in different healthcare settings.

The application of balancing in most cases improved specificity at the expense of sensitivity, indicating that while models became better at identifying the minority class (short ED LOS), they were less effective at detecting the majority class (long ED LOS). This trade-off suggests that the some balancing method may not be suitable for all clinical contexts, particularly where high sensitivity is crucial (54).

Furthermore, some rules, such as Long Rule 1, may exhibit a high overall coverage but a lower relative coverage, indicating a relatively high rate of misclassifications within the predicted category. For instance, Long rule 1 has an overall coverage of 74.78%, but its relative coverage for short stays is 23.49% and for long stays 51.28%, suggesting that while the rule applies to a large portion of the entire dataset, it may sometimes not accurately distinguish between short and long stays.

In addition, an argument-based framework that integrates learning and reasoning (53) will be explored. In this framework, knowledge is represented through object-level arguments that involve constructing arguments that link a set of premises (conditions describing a scenario) to the argument's claim (the desired outcome). Two types of arguments are generated: objectlevel and priority arguments. Object-level arguments are factual claims and may support contradictory assertions, leading to opposing arguments. However, priority arguments establish a local preference between these arguments, determining their relative strength and intensifying the conflict between them. This framework will be exploited by analyzing the conflict between long and short rules.

Future research will also investigate the application of these techniques in diverse healthcare settings, evaluate the generalizability of generated rules, and improve rule extraction methods to better handle more complex cases and models.

6 Conclusion

This study demonstrates a robust machine learning-based approach to predicting Emergency Department Length of Stay (ED LOS), with Gradient Boosting (GB) emerging as the best performer in our analysis. Key predictors for ED LOS included Triage Acuity, Elixhauser Comorbidity Index, Arrival Methods, and Patient Medication History. Our use of multiple data balancing techniques, including SMOTETomek, effectively addressed class imbalance, but a trade-off between sensitivity and specificity persisted across models.

This work made an important contribution by extracting rules from the GB model using the TE2Rules algorithm. This process enabled us to create interpretable rules to aid clinical decisionmaking. These rules capture important relationships, such as patient age, triage acuity, and arrival conditions, which improves model transparency. Our findings indicate that combining predictive accuracy and interpretability via rule extraction can improve resource management and decision-making in emergency departments.

While the models produced promising results, there is still room for improvement in achieving a better balance of performance metrics. Future efforts will center on improving rule extraction processes, increasing model generalizability, and ensuring adaptability across various healthcare settings.

Data availability statement

Publicly available datasets were analyzed in this study. This data can be found here: Johnson et al. (22). In order to aid reproducibility, the code used in this study is available at the following GitHub repository: https://github.com/waziz786/ MIMIC-IV-ED-CODE.

Ethics statement

Ethical approval was not required for the study involving humans in accordance with the local legislation and institutional

requirements. Written informed consent to participate in this study was not required from the participants or the participants' legal guardians/next of kin in accordance with the national legislation and the institutional requirements.

Author contributions

WS: Conceptualization, Data curation, Formal Analysis, Investigation, Methodology, Software, Writing - original draft, Writing - review & editing. CS: Writing - review & editing. AN: Writing - review & editing, Conceptualization, Methodology. ZA: Writing - review & editing, Writing - original draft. IC: Writing - original draft, Writing - review & editing. LP: Writing - review & editing, Conceptualization, Methodology. AV: Conceptualization, Methodology, Writing - original draft. TK: Conceptualization, Methodology, Writing - review & editing. EK: Conceptualization, Writing - review & editing. AK: Conceptualization, Writing - review & editing. MSP: Conceptualization, Writing - review & editing. ASP: Conceptualization, Writing - review & editing. CSP: Conceptualization, Formal Analysis, Investigation, Project administration, Supervision, Writing - review & editing.

Funding

The authors declare financial support was received for the research, authorship, and/or publication of this article. This work was partly funded by the European Union Recovery and Resilience Facility of the NextGenerationEU instrument, through

References

1. Canadian Association of Emergency Physicians, National Emergency Nurses Affiliation. Access to acute care in the setting of emergency department overcrowding. *Can J Emerg Med.* (2003) 5(2):81–6. doi: 10.1017/ S1481803500008204

2. Jung HM, Kim MJ, Kim JH, Park YS, Chung HS, Chung SP, et al. The effect of overcrowding in emergency departments on the admission rate according to the emergency triage level. *PLoS One*. (2021) 16(2):e0247042. doi: 10.1371/journal.pone. 0247042

3. Hunter-Zinck HS, Peck JS, Strout TD, Gaehde SA. Predicting emergency department orders with multilabel machine learning techniques and simulating effects on length of stay. *J Am Med Inform Assoc.* (2019) 26(12):1427-36. doi: 10. 1093/jamia/ocz171

4. Shafaf N, Malek H. Applications of machine learning approaches in emergency medicine; a review article. *Arch Acad Emerg Med.* (2019) 7(1):34.

5. Raita Y, Goto T, Faridi MK, Brown DF, Camargo CA Jr, Hasegawa K. Emergency department triage prediction of clinical outcomes using machine learning models. *Crit Care.* (2019) 23(1):64. doi: 10.1186/s13054-019-2351-7

6. Xie F, Ong MEH, Liew JNMH, Tan KBK, Ho AFW, Nadarajan GD, et al. Development and assessment of an interpretable machine learning triage tool for estimating mortality after emergency admissions. *JAMA Network Open.* (2021) 4(8): e2118467. doi: 10.1001/jamanetworkopen.2021.18467

7. Xie F, Zhou J, Lee JW, Tan M, Li S, Rajnthern LS, et al. Benchmarking emergency department prediction models with machine learning and public electronic health records. *Sci Data*. (2022) 9(1):658. doi: 10.1038/s41597-022-01782-9

8. Hassija V, Chamola V, Mahapatra A, Singal A, Goel D, Huang K, et al. Interpreting black-box models: a review on explainable artificial intelligence. *Cognit Comput.* (2023) 16:45–74. doi: 10.1007/s12559-023-10179-8 the Research and Innovation Foundation of Cyprus (Project: CODEVELOP-ICT-HEALTH/0322/0071, Hospital Transformation through Artificial Intelligence – HOSPAITAL).

Conflict of interest

Authors ZA and IC were employed by company 3AHealth.

The remaining authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

The authors declared that they were an editorial board member of Frontiers, at the time of submission. This had no impact on the peer review process and the final decision.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Supplementary material

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fdgth.2024. 1498939/full#supplementary-material

9. Srinivasu PN, Sandhya N, Jhaveri RH, Raut R. From blackbox to explainable AI in healthcare: existing tools and case studies. *Mobile Inf Syst.* (2022) 2022:1–20. doi: 10. 1155/2022/8167821

10. Al-A'araji NH, Al-Mamory SO, Al-Shakarchi AH. The impact of rule evaluation metrics as a conflict resolution strategy. *Commun Comput Inf Sci.* (2020) 1183:105–27. doi: 10.1007/978-3-030-55340-1_8

11. Zavala-Díaz J, Olivares-Rojas JC, Gutiérrez-Gnecchi JA, Alcaraz-Chávez JE, Reyes-Archundia E, Téllez-Anguiano AC, et al. Clinical notes classification system for automated identification of diabetic patients: hybrid approach integrating rules, information extraction and machine learning. *J Intell Fuzzy Syst.* (2024) 47:1–11. doi: 10.3233/jifs-219375

12. Alsinglawi B, Alshari O, Alorjani M, Mubin O, Alnajjar F, Novoa M, et al. An explainable machine learning framework for lung cancer hospital length of stay prediction. *Sci Rep.* (2022) 12(1):607. doi: 10.1038/s41598-021-04608-7

13. Karajeh A, Eskicioglu R. Classifying emergency patients into fast-track and complex cases using machine learning. *Int J Artif Intell Appl.* (2024) 15(3):53–65. https://papers.ssrn.com/sol3/papers.cfm?abstract_id=4863348

14. Zebin T, Rezvy S, Chaussalet TJ. A deep learning approach for length of stay prediction in clinical settings from medical records. In 2019 IEEE Conference on Computational Intelligence in Bioinformatics and Computational Biology (CIBCB); 2019 Jul 09-11; Siena, Italy. Piscataway, NJ: IEEE Xplore (2019) 15(3):1-5. Available online at: https://ieeexplore.ieee.org/ document/8791477.

15. Curiati PK, Gil-Junior LA, Morinaga CV, Ganem F, Curiati JAE, Avelino-Silva TJ. Predicting hospital admission and prolonged length of stay in older adults in the emergency department: the PRO-AGE scoring system. *Ann Emerg Med.* (2020) 76(3):255–65. doi: 10.1016/j.annemergmed.2020.01.010 16. Bopche R, Gustad LT, Afset JE, Birgitta E, Damås JK, Øystein N. In-hospital mortality, readmission, and prolonged length of stay risk prediction leveraging historical electronic patient records. *JAMIA Open.* (2024) 7(3):00ae074. doi: 10. 1093/jamiaopen/00ae074

17. Etu EE, Monplaisir L, Arslanturk S, Masoud S, Aguwa C, Markevych I, et al. Prediction of length of stay in the emergency department for COVID-19 patients: a machine learning approach. *IEEE Access.* (2022) 10:42243–51. doi: 10.1109/ACCESS.2022.3168045

18. Chang YH, Shih HM, Wu JE, Huang FW, Chen WK, Chen DM, et al. Machine learning–based triage to identify low-severity patients with a short discharge length of stay in emergency department. *BMC Emerg Med.* (2022) 22:88. doi: 10.1186/s12873-022-00632-6

19. Rahman MA, Honan B, Glanville T, Hough P, Walker K. Using data mining to predict emergency department length of stay greater than 4 h: derivation and singlesite validation of a decision tree algorithm. *Emerg Med Australas*. (2019) 32(3):416–21. doi: 10.1111/1742-6723.13421

20. Gill SD, Lane SE, Sheridan M, Ellis E, Smith D, Stella J. Why do "fast track" patients stay more than four hours in the emergency department? An investigation of factors that predict length of stay. *Emerg Med Australas.* (2018) 30(5):641–7. doi: 10.1111/1742-6723.12964

21. Silva AR, Morinaga CV, Avelino-Silva TJ, Pedro KC. Unplanned intensive care unit admission in hospitalized older patients: association with a geriatric vulnerability score. *J Am Geriatr Soc.* (2022) 70(11):3321–4. doi: 10.1111/jgs.17944

22. Johnson AEW, Bulgarelli L, Shen L, Gayles A, Shammout A, Horng S, et al. MIMIC-IV, a freely accessible electronic health record dataset. *Sci Data.* (2023) 10(1). doi: 10.1038/s41597-022-01899-x

23. Zhou T (2021). Addressing deficiencies from missing data in electronic health records. Mitedu. Available online at: https://dspace.mit.edu/handle/1721.1/139587 (accessed June 25, 2021).

24. Choi S, Kim G, Kwon JE, Lee HK. Sequential anomaly detection for continuous prediction of unexpected ICU admission from emergency department. *IEEE Access*. (2024) 12:96304–18. doi: 10.1109/ACCESS.2024.3426675

25. Wang S, McDermott MBA, Chauhan G, Ghassemi M, Hughes MC, Naumann T, et al. MIMIC-extract: a data extraction, preprocessing, and representation pipeline for MIMIC-III. *ACM Chil.* (2020) 20:222–35. doi: 10.1145/3368555.3384469

26. Harutyunyan H. YerevaNN/Mimic3-benchmarks. GitHub. Available online at: github.com/YerevaNN/mimic3-benchmarks (accessed March 14, 2024).

27. Liu FT, Ting KM, Zhou ZH. Isolation forest. In: 2008 8th IEEE International Conference on Data Mining, Pisa, 15-19 December 2008. Scientific Research Publishing (2008). p. 413–22. Available online at: https://www.scirp.org/reference/referencespapers?referenceid=3054470.

28. Alghushairy O, Alsini R, Soule T, Ma X. A review of local outlier factor algorithms for outlier detection in big data streams. *Big Data Cogn Comput.* (2020) 5(1):1. doi: 10.3390/bdcc5010001

29. Prakash S, Singh S, Ashutosh M. Bridging data gaps: a comparative study of different imputation methods for numeric datasets. In: 2024 International Conference on Data Science and Network Security (ICDSNS). (2024). p. 1–7. Available online at: https://ieeexplore.ieee.org/abstract/document/10691111.

30. Beaulieu-Jones BK, Moore JH. Missing data imputation in the electronic health record using deeply learned autoencoders. *Biocomputing*. (2016) 2017:207–18. doi: 10. 1142/9789813207813_0021

31. Andersson J, Nordgren L, Cheng J, Nilsson U, Kurland L. Long emergency department length of stay: a concept analysis. *Int Emerg Nurs.* (2020) 53:100930. doi: 10.1016/j.ienj.2020.100930

32. Judge TP. Reforming emergency care and ambulance services. *Emerg Med J.* (2004) 21(1):4. doi: 10.1136/emj.2003.008003

33. imblearn: toolbox for imbalanced dataset in machine learning. PyPI. Available online at: https://pypi.org/project/imblearn/. (Accessed December 11, 2024).

34. Elreedy D, Atiya AF. A novel distribution analysis for SMOTE oversampling method in handling class imbalance. *Lect Notes Comput Sci.* (2019) 11538:236-48. doi: 10.1007/978-3-030-22744-9_31

35. He H, Yang B, Garcia EA, Shutao L. ADASYN: Adaptive synthetic sampling approach for imbalanced learning. In 2008 IEEE International Joint Conference on Neural Networks (IEEE World Congress on Computational Intelligence; 2008 Jun 01-08; Hong Kong. New York: IEEE Xplore (2008). p. 1322-8.

36. Tomek I. Two modifications of CNN. IEEE Trans Syst Man Cybern. (1976) SMC-6(11):769-72. doi: 10.1109/TSMC.1976.5409182

37. Hairani H, Priyanto D. A new approach of hybrid sampling SMOTE and ENN to the accuracy of machine learning methods on unbalanced diabetes disease data. *Int J Adv Comput Sci Appl.* (2023) 14(8):585–90. doi: 10.14569/IJACSA.2023. 0140864

38. Menendez ME, Neuhaus V, van Dijk CN, Ring D. The elixhauser comorbidity method outperforms the charlson index in predicting inpatient death after orthopaedic surgery. *Clin Orthop Relat Res.* (2014) 472(9):2878–86. doi: 10.1007/s11999-014-3686-7

39. pandas.get_dummies—pandas 1.2.4 documentation. Available online at: https://pandas.pydata.org/docs/reference/api/pandas.get_dummies.html (Accessed December 12, 2024).

40. Auret L, Aldrich C. Empirical comparison of tree ensemble variable importance measures. *Chemometr Intell Lab Syst.* (2011) 105(2):157–70. doi: 10.1016/j.chemolab. 2010.12.004

41. Chowdhury MZI, Turin TC. Variable selection strategies and its importance in clinical prediction modelling. *Family Med Community Health.* (2020) 8(1). doi: 10. 1136/fmch-2019-000262

42. Scikit-learn. Scikit-learn: machine learning in python. Scikit-Learn.org (2019). Available online at: scikit-learn.org/stable/ (Accessed December 12, 2024).

43. Chollet FK. GitHub repository (2015). Available online at: https://github.com/ fchollet/keras (Accessed December 12, 2024).

44. Marcot BG, Hanea AM. What is an optimal value of k in k-fold cross-validation in discrete Bayesian network analysis?. *Comput Stat.* (2020) 36:2009–31. doi: 10.1007/s00180-020-00999-9

45. Japkowicz N, Shah M. Performance evaluation in machine learning. In: El Naqa I, Li R, Murphy M, editors. *Machine Learning in Radiation Oncology*. Cham: Springer (2015).

46. Lal GR, Chen X, Mithal V. TE2Rules: explaining tree ensembles using rules. arXiv [Preprint]. arxiv.org/abs/2206.14359 (2022) (accessed 20 July 2024).

47. Lal GR. Te2rules: python library to explain tree ensembles using rules. PyPI. Available online at: pypi.org/project/te2rules/ (Accessed December 12, 2024).

48. Wróbel Ł, Sikora M, Michalak M. Rule quality measures settings in classification, regression and survival rule induction – an empirical approach. *Fundam Inform*. (2016) 149(4):419–49. doi: 10.3233/FI-2016-1455

49. Bentéjac C, Csörgö A, Martínez-Muñoz G. A comparative analysis of gradient boosting algorithms. *Artif Intell Rev.* (2020) 54:1937–67. doi: 10.1007/ s10462-020-09896-5

50. Nicolaou A, Charithea S, Sulaiman WA, Antoniou Z, Lakis P, Vavlitou A, et al. An overview of explainable AI studies in the prediction of sepsis onset and sepsis mortality. *Stud Health Technol Inform.* (2024) 316:808–12. doi: 10.3233/SHTI240534

51. Kübra A, Mehmet K. Examining patients length of stay estimation with explainable artificial intelligence methods. In Information Systems Engineering and Management Proceedings of the Fifth International Conference on Emerging Trends in Mathematical Sciences & Computing (IEMSC-24). Cham, Switzerland: Springer Nature (2024). p. 296-311.

52. Aziz W, Nicalaou A, Stylianides C, Panayides A, Kakas A, Kyriacou E, et al. Emergency department length of stay classification based on ensemble methods and rule extraction. *Stud Health Technol Inform.* (2024) 316:1812-6. doi: 10.3233/SHTI240783

53. Kakas AC, Moraitis P, Spanoudakis NI. GORGIAS: applying argumentation. Argument Comput. (2019) 10(1):55–81. doi: 10.3233/AAC-181006

54. Fernández A, García S, Galar M, Prati RC, Krawczyk B, Herrera F. Learning From Imbalanced Data Sets. Cham, Switzerland: Springer (2018).